Author's response to reviews

Title: Reversal of childhood idiopathic scoliosis in an adult, without surgery: a case report and literature review

Authors:

William J Brooks (wjbdo@wjbrooksdo.com)
Elizabeth A Krupinski (krupinski@radiology.arizona.edu)
Martha C Hawes (mhawes@u.arizona.edu)

Version: 8 Date: 1 December 2009

Author's response to reviews:

Our responses in capital letters:

Dear Prof Hawes,

We are delighted to confirm that Scoliosis will publish your manuscript (above) pending formatting requirements. We ask that you make the following changes to the format of your manuscript so that online production process of your manuscript is straightforward. After making these changes, please resubmit a new version of your paper to the submission system. The instructions to submit your revised manuscript are available below.

Formatting changes to be made:

-----------------------------

Major revisions (We require the author to make these changes)

Competing interests: Please include a `Competing interests' section between the Conclusions and Authors' contributions. If there are none to declare, please write `The authors declare that they have no competing interests'.

Please see here for a full list of questions to consider when writing your competing interests statement-

DONE, AS REQUESTED.

Tables: tables must be created using a Tables tool, and should not be linked files, figures, or generated with tabbed text. Tables must be divided into rows and columns. We recommend the MS Word auto-format template: simple 1. Please do not include vertical lines in your tables.

DONE, AS REQUESTED.
The table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key), which should appear below the table.

DONE, AS REQUESTED.

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

Affiliations: Please remember to include the country in the affiliation details.

DONE, AS REQUESTED.

References: the reference list should contain up to 30 authors' names for each citation. The term 'et al.' should not be used.

DONE, AS REQUESTED.

Blank page: Please remove all blank pages from the manuscript.

DONE, AS REQUESTED.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.

DONE, AS REQUESTED.

Please also ensure that all figure citations match the figure titles. Currently some citations are for `Fig', but they are titled `Figure'.

DONE, AS REQUESTED.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

DONE, AS REQUESTED.

Thank you